

RESEARCH ARTICLE

Leishmaniasis and tumor necrosis factor alpha antagonists in the Mediterranean basin. A switch in clinical expression

Pau Bosch-Nicolau¹, Maria Ubals², Fernando Salvador¹, Adrián Sánchez-Montalvá¹, Gloria Aparicio², Alba Erra³, Pablo Martinez de Salazar⁴, Elena Sulleiro⁴, Israel Molina^{1*}

1 Department of Infectious Diseases, Hospital Universitari Vall d'Hebron, PROSICS Barcelona, Universitat Autònoma de Barcelona, Spain, 2 Department of Dermatology, Hospital Universitari Vall d'Hebron, Universitat Autònoma de Barcelona, Spain, 3 Department of Rheumatology, Hospital Universitari Vall d'Hebron, Universitat Autònoma de Barcelona, Spain, 4 Department of Clinical Microbiology, Hospital Universitari Vall d'Hebron, PROSICS Barcelona, Universitat Autònoma de Barcelona, Spain

* imolina@vhebron.net

Abstract

Background

Tumor necrosis factor alpha (TNF- α) blockers are recognized as a risk factor for reactivation of granulomatous infections. Leishmaniasis has been associated with the use of these drugs, although few cases have been reported.

Methodology

We performed a retrospective observational study including patients with confirmed leishmaniasis acquired in the Mediterranean basin that were under TNF- α blockers therapy at the moment of the diagnosis. Patients diagnosed in our hospital from 2008 to 2018 were included. Moreover, a systematic review of the literature was performed and cases fulfilling the inclusion criteria were also included.

Principal findings

Forty-nine patients were analyzed including nine cases from our series. Twenty-seven (55.1%) cases were male and median age was 55 years. Twenty-five (51%) patients were under infliximab treatment, 20 (40.8%) were receiving adalimumab, 2 (4.1%) etanercept, one (2%) golimumab and one (2%) a non-specified TNF- α blocker. Regarding clinical presentation, 28 (57.1%) presented as cutaneous leishmaniasis (CL), 16 (32.6%) as visceral leishmaniasis (VL) and 5 (10.2%) as mucocutaneous leishmaniasis (MCL). All VL and MCL patients were treated with systemic therapies. Among CL patients, 13 (46.4%) were treated with a systemic drug (11 received L-AmB, one intramuscular antimonials and one miltefosine) while 14 (50%) patients were given local treatment (13 received intralesional pentavalent antimonials, and one excisional surgery). TNF- α blockers were interrupted in 32 patients (65.3%). After treatment 5 patients (10.2%) relapsed. Four patients with a CL (3



GOPEN ACCESS

Citation: Bosch-Nicolau P, Ubals M, Salvador F, Sánchez-Montalvá A, Aparicio G, Erra A, et al. (2019) Leishmaniasis and tumor necrosis factor alpha antagonists in the Mediterranean basin. A switch in clinical expression. PLoS Negl Trop Dis 13(8): e0007708. https://doi.org/10.1371/journal. pntd.0007708

Editor: Abhay R Satoskar, Ohio State University, UNITED STATES

Received: May 13, 2019

Accepted: August 15, 2019

Published: August 30, 2019

Copyright: © 2019 Bosch-Nicolau et al. This is an open access article distributed under the terms of the Creative Commons Attribution License, which permits unrestricted use, distribution, and reproduction in any medium, provided the original author and source are credited.

Data Availability Statement: All relevant data are within the manuscript and its Supporting Information files.

Funding: The author received no specific funding for this work.

Competing interests: The authors have declared that no competing interests exist.

initially treated with local therapy maintaining TNF- α blockers and one treated with miltefosine) and one patient with VL treated with L-AmB maintaining TNF- α blockers.

Conclusions

This data supports the assumption that the blockage of TNF- α modifies clinical expression of leishmaniasis in endemic population modulating the expression of the disease leading to atypical presentations. According to the cases reported, the best treatment strategy would be a systemic drug and the discontinuation of the TNF- α blockers therapy until clinical resolution.

Author summary

Tumor necrosis factor alpha (TNF- α) blockers are widely used in numerous inflammatory diseases such rheumatoid arthritis, psoriasis or inflammatory bowel diseases. They have been recognized as a risk factor for reactivation of granulomatous infections. Although few cases have been reported, Leishmaniasis has been associated with the use of these drugs. *Leishmania infantum* is the main causative agent of leishmaniasis in Southern Europe and is prone to produce the visceral form. However, TNF- α has been implicated in the initial events of the infection mediating the disease expression. In our series, we have observed a surprisingly high proportion of cutaneous form (32.6%) and muco-cutaneous form (10.2%). Clinical outcome observed in this series is also unusual. Four cases (14.3%) with cutaneous leishmaniasis, one patient who maintained TNF- α blockers therapy relapsed despite etiological treatment. This data supports the assumption that the blockage of TNF- α modifies clinical expression of leishmaniasis leading to atypical presentations. According to the cases reported we proposed as best treatment strategy a systemic drug and the discontinuation of the TNF- α blockers therapy until clinical resolution.

Introduction

Tumor Necrosis Factor- α (TNF- α) is a crucial cytokine in the inflammatory cascade by activating the type 1 T helper (Th1) immune response, enhancing the activity of the macrophages and essential for the formation and maintenance of granulomas [1].

Since TNF- α has been implicated in numerous immune-mediated disorders, the blockage of this cytokine has been studied as a therapeutic strategy against such diseases. Nowadays, the anti-TNF based therapy is widely used and approved for the treatment of chronic inflammatory conditions as rheumatoid arthritis, polyarticular juvenile idiopathic arthritis, plaque psoriasis and psoriatic arthritis, ankylosing spondylitis and inflammatory bowel diseases [2]. The first approved TNF- α blocker was etanercept (Enbrel) in May 1998 followed by infliximab (Remicade) in November 1999, adalimumab (Humira) in December 2002, certolizumab (Cinzia) in April 2008 and golimumab (Simponi) in April 2009.

Since their first use, the TNF- α blockers were recognized as a risk factor for reactivation of granulomatous infections such as tuberculosis, intracellular infections such as salmonellosis or listeriosis and other opportunistic fungal or viral infections [3].

Leishmaniasis is a parasitic granulomatous infection and it is endemic to South America, South Asia, Africa and South Europe. The protozoon is an obligate intracellular parasite of mononuclear phagocytic system cells. The clinical spectrum of leishmaniasis comprises subclinical (asymptomatic), localized (cutaneous) and disseminated infection (cutaneous, mucosal and visceral). Its clinical expression is determined on one hand by the species and zimodeme of the parasite and on the other hand by host factors and immune response [4].

Leishmaniasis has been associated with the use of TNF- α blockers, but only few cases have been reported in the literature, mainly in the Mediterranean basin [5,6]. We report nine more cases related to the use of TNF- α blockers and systematically review the published cases acquired in the Mediterranean basin. We also analyze their clinical presentation and discuss the relationship with immunomodulatory therapy. Finally, a therapeutic approach is discussed.

Methods

We carried out a retrospective observational study including patients with a diagnosis of leishmaniasis in its different forms. All patients were under TNF- α blockers and were diagnosed in our center between 2008 and 2018. We also performed MEDLINE search using the terms *Leishmania, leishmaniasis, TNF-\alpha inhibitors, TNF-\alpha blockers, anti-TNF-\alpha, adalimumab, infliximab, etanercept, certolizumab and golimumab*. No language or time restrictions were applied. A manual search of the references of the selected manuscript was also performed.

In order to be included in the analysis, cases had to be diagnosed based on direct observation of amastigotes and/or positive *Leishmania* polymerase chain reaction (qPCR) of blood, bone marrow or skin samples.

DNA extraction from blood and bone marrow samples was carried out from 110 µl with silica–membrane technology (NucliSens easyMAG. Biomerieux. France) and eluted in 110 µl according to the manufacturer's instructions. The extraction protocol from skin biopsy was made in a Magcore Compact (RBC Bioscience. Taiwan) and eluted in 100 µl according to the manufacturer's instructions. A duplex qPCR targeted to kinetoplast minicircle DNA *of Leishmania sp* and human RNase P gene (Taq Man Human RNase P detection reagent; Applied Biosystems) were performed using the primers and probe described previously [7]. Cycling conditions were a first step of 15 minutes at 95°C followed by 45 cycles at 95°C for 15 seconds and 55°C for 1 minute. Amplifications were carried out in a CFX Real-Time PCR detection system (Bio-Rad, Hercules, CA). *Leishmania infantum* identification was performed by restriction fragment length polymorphism of the internal transcribed spacer regions (ITS-RFLP) in a reference laboratory.

Patients were classified based on the clinical form as follows: VL when compatible signs and symptoms and blood or bone marrow samples had a positive culture or *Leishmania* polymerase chain reaction (qPCR). MCL when compatible signs and symptoms and mucose samples had a positive culture or *Leishmania* qPCR. CL when compatible signs and symptoms and cutaneous samples had a positive culture *Leishmania* qPCR. We assessed the patients' complete medical records: demographic data, underlying disease, TNF- α blockers therapy, specific treatment and outcomes were recorded. Cure was defined as an absence of clinical signs after a minimum follow-up of 1 year after leishmanicidal treatment discontinuation. Unrelated death was established when a death with no relation to the infection or its treatment discontinuation during the first year of follow-up unless relapse occurring after this period can be microbiologically demonstrated. Those cases without relevant information or not well identified were excluded from analysis.

Statistical analysis

Categorical variables are expressed as percentages, and numerical data as the mean \pm SD for variables with a normal distribution or the median (IQR) for those with a skewed distribution. Categorical variables were compared with the chi-square test or Fisher exact test, and continuous variables with the Student *t* or the Mann-Whitney *U* test, depending on distribution. All statistical tests were 2-tailed, and significance was set at P < .05. Statistical analyses were performed using SPSS version 20.0 (SPSS, Inc., Chicago, IL, USA).

Ethics statement

Due to its retrospective design, oral consent was obtained by phone contact from the included patients. The study was approved by the Ethics Committee of Vall d'Hebron Research Institute.

Results

A total of 33 publications were retrieved in our search including forty cases that fulfilled inclusion criteria [8–39]. Thus, a total of 49 cases were analyzed including our 9 cases (Table 1). Twenty-seven (55.1%) cases were male and the median age was 55 (range 7–80) years. Twenty-five (51%) patients were under infliximab treatment at the moment of leishmaniasis diagnosis, twenty (40.8%) were receiving adalimumab, two (4.1%) were receiving etanercept, one (2%) golimumab and another one (2%) received a non-specified TNF- α blocker. From greater to lesser frequency, the underlying disease was psoriatic arthritis in twelve (24.5%) cases, rheumatoid arthritis in twelve (24.5%) cases, ankylosing spondylitis in nine (18.4%) cases, Crohn's diseases in five (10.2%) cases, plaque psoriasis in five (10.2%) cases and rheumatoid arthritis with psoriasis, ulcerative colitis, juvenile idiopathic arthritis, giant cell arthritis, seronegative arthritis and *folliculitis decalvans* in one case each other. All patients were diagnosed in European hospitals and probable place of infection was Spain in thirty-three (67.3%) cases, followed by Greece in five (10.5%) cases, Italy in four (8.2%) cases, France in two (4.1%) cases, Malta in two (4.1%) cases, Algeria in two (4.1%) cases and Turkey in one (2%) case.

Regarding the clinical presentation, CL was the most frequent form (28 patients, 57.1%), eighteen cases presenting as a solitary ulcerative lesion and ten cases including one attended in our hospital who presented multifocal lesions (Fig 1). Sixteen (32.6%) patients presented VL and five (10.2%) patients had MCL, three in the nasal cavity (Fig 2), one as hyperplasic lesions around perianal mucosa and one as an infiltrative tumor involving upper lip, hard palate and nasal septum. Bone marrow aspirate was performed in one patient with MCL and one patient with CL. Although no amastigotes forms were observed, the patient with MCL had a positive *Leishmania* RT-PCR on the bone marrow sample. Identification to the level of species could be performed in eighteen patients. *L. infantum* was identified in seventeen cases and one was reported as *L. donovani complex*.

All MCL and VL patients were treated with a systemic therapy. Fifteen (71.4%) of them were treated with liposomal amphotericine B (L-AmB), including one patient treated in combination with intralesional pentavalent antimonials, three (14.3%) patients were treated with parenteral pentavalent antimonials and one (4.8%) patient with miltefosine. Among CL cases, thirteen (46.4%) patients received a systemic treatment; eleven were given L-AmB, one intramuscular antimonials and one miltefosine. Local treatment was given to fourteen (50%) patients, thirteen received intralesional pentavalent antimonials, one combined with surgery and another with cryotherapy and one patient was treated with surgical excision of the lesion. Only one patient defaulted and did not receive any treatment.

Comment			Death was due to respiratory superinfection					Relapsed 26months after L-Amb cessation with a MCL form.						Episode considered as visceralization of a CL form	(Continued)
Outcome	Cure	Cure	Dead	Cure	Cure	Cure	Cure	Relapse	Cure	Cure	Cure	Cure	Cure	Cure	
Treatment	Parenteral Sb	L-AmB 15.4mg/Kg Anti-TNF Stopped	L-AmB 50mg/ Kg	L-AmB 18mg/ Kg Anti-TNF Stopped	L.Amb 18mg/ Kg Anti-TNF Stopped	L-Amb Anti-TNF Stopped	Parenteral Sb Anti-TNF Stopped	L-Amb 24mg/ Kg	L-Amb 21mg/ Kg	L-AmB 21mg/ Kg Anti-TNF Stopped	L-AmB 30mg/ Kg Anti-TNF Stopped	L-AmB Anti-TNF Stopped	L-Amb Anti-TNF Stopped	L-Amb 30mg/ Kg Intralesional Sb Anti-TNF Stopped	
Diagnostic	DO+	DO+ PCR+	+00+	DO+ PCR+	PCR+	DO+	+00+	DO+	DO+ PCR+	DO+	D0+	+00+	D0+	DO+ PCR+	
Sample	Bone Marrow	Bone Marrow	Bone Marrow	Bone Marrow	Bone Marrow	Bone Marrow	Bone Marrow	Bone Marrow	Bone Marrow	Bone marrow Duodenal biopsy	Bone Marrow	Bone Marrow	Bone Marrow	Bone Marrow Skin biopsy	
Specie	Leishmania sp.	L. infantum MON-1	Leishmania sp.	Leishmania sp.	Leishmania sp.	Leishmania sp.	L. donovani complex	L. infantum	Leishmania sp.	Leishmania sp.	Leishmania sp.	Leishmania sp.	Leishmania sp.	L. infantum	
Number of lesions			1			1	1					1	1	,	
Clinical form	ΤΛ	TA	ΤΛ	TA	TA	ΤΛ	ΤΛ	TA	ΓΛ	ΤΛ	TA	ΛΓ	ΛΓ	VL/CL	
Disease	PA	RA	RA	RA	PA	PA	AS	JIA	GCA	RA	RA	RA	RA	PA	
Sex-Age	M-55	F-53	F-60	F-69	M-45	M-63	M-55	F-7	F-71	F-72	M-60	F-77	F-71	F-74	
Other IS	,	Azatioprine Corticosteroids	Cyclosporin Corticosteroids	MTX Corticosteroids	MTX 12.5mg/ w Pred 7.5g/d	MTX 7.5mg/w AZA (different times)	MTX Corticosteroids	MTX 10mg/w	MTX 10mg/w Corticosteroids	MTX 15mg/w Pred 5-10 mg/ d	MTX 20 mg/w Pred 5-10 mg/ d	1	MTX 10mg/w	XTM	
Anti TNF	Infliximab	Infliximab	Etanercept	Adalimumab	Infliximab	Infliximab	Infliximab	Infliximab	Infliximab	Infliximab	Adalimumab	Infliximab	Adalimumab	Adalimumab	
Country (Region)	Spain (Catalonia)	France (Languedoc- Rousillon)	Greece (Central Macedonia)	Italy (Liguria)	Greece (Atica)	Italy (Emilia Romagna)	Spain (Catalonia)	France (Languedoc- Roussillon)	Greece (Atica)	Spain (Valencian Comunity)	Spain (Catalonia)	Greece (Atica)	Italy (Sardinia)	Malta	
Author, year and reference	Romaní- Costa 2004 (8)	Fabre 2005 (9)	Bagalas 2006 (10)	Bassetti 2006 (11)	Tektonidou 2008 (12)	De Leonardis 2008 (13)	Garcia- Vidal 2009 (14)	Jeziorski 2009 (15)	Xynos 2009 (16)	Moreno 2010 (17)	Moltó 2010 (18)	Kritikos 2010 (19)	Erre 2010 (20)	Khan 2010 (21)	
Patient Number	1	2	3	4	ъ	و	7	×	6	10	11	12	13	14	

PLOS | NEGLECTED TROPICAL DISEASES

	Comment						ath was due to al arrhythmia side effect of antimony	Episode onsidered as a activation of a previous CL curred 4 years before	C form treated th L-AmB/ IM b. Anti-TNF opped Finally cured	form treated h intralesional b. Anti-TNF opped Finally cured		esented several sted lesions on face	
	Outcome	Cure	Cure	Cure	Cure	Cure	Dead De fat as	Cure co	Relapse M(win St St	Relapse C wit St St	Cure	Cure Pre	Cure
	Treatment	L-AmB 21mg/ Kg Anti-TNF Stopped	L.AmB 40mg/ Kg Anti-TNF Stopped	L-AmB/Sb Anti-TNF Stopped	L-AmB 15mg/ Kg Anti-TNF Stopped	Miltefosine	Intramuscular MgA	L-AmB 40mg/ Kg Anti-TNF Stopped	Intralesional Sb Surgery	Miltefosine Anti-TNF Stopped	L-AmB 21mg/ Kg Anti-TNF Stopped	L-AmB 21mg/ Kg Anti-TNF Stopped	Intralesional MgA Anti-TNF Stopped
	Diagnostic	DO+ PCR +	DO+ PCR+	DO+ Culture +	DO+ Culture+	PCR+	DO+ PCR+	DO +(biopsy) PCR +	DO+ PCR+	DO+ PCR+ Culture +	PCR+	D0+	DO+ PCR+ Culture +
	Sample	Bone Marrow	Bone Marrow Spleen Biopsy	Mucosa biopsy	Mucosal biopsy	Skin biopsy	Perianal Biopsy	Mucosal biopsy Bone marrow	Skin Biopsy	Skin Biopsy	Skin Biopsy	Skin Biopsy	Skin Biopsy
	Specie	L. infantum	L. infantum	L. Infantum MON-183	L. infantum	L. infantum	L. infantum	L. Infantum	L. infantum	L. Infantum MON 1	L. infantum	Leishmania sp.	L. infantum MON-1
	Number of lesions			-	-	-		-	-	-	ŝ	Several	ŝ
	Clinical form	TA	TA	MCL	MCL	MCL	MCL	MCL/ VL	Ъ	C	C	C	CL
	Disease	PA	RA	RA	AS	PA	CD	PA	RA	AS	AS	AS	AS
	Sex-Age	F-80	F-69	F-66	M-49	M-53	M-44	M-53	F-56	M-31	F-51	M-55	M-50
	Other IS	MTX 20mg/w	MTX 20mg/w	MTX 15mg/w Pred 2.5mg/d	MTX 10mg/w M-pred 4mg/d	1	MTX 15mg/w		MTX 15mg/w Pred 3.75 mg/d	1	MTX	MTX 10mg/w	,
	Anti TNF	Etanercept	Adalimumab	Infliximab	Adalimumab	Adalimumab	Infliximab	Infliximab	Infliximab followed Adalimumab	Infliximab	Adalimumab	Infliximab	Infliximab
	Country (Region)	Spain (Valencian Comunity)	Spain (Andalusia)	Spain (Andalusia)	Italy (Tuscany)	Spain (Balearic Islands)	Turkey/ Croatia	Spain (Catalonia)	Spain (Catalonia)	Spain (Balearic Islands)	Algeria	Greece (Atica)	Algeria / France (Languedoc- Rousillon)
(Continued)	Author, year and reference	Besada 2013 (22)	Berry 2013 (23)	García- Castro 2011 (24)	García- González 2012 (25)	Neumayr 2013 (26)	Juzlova 2014 (27)	Bosch- Nicolau 2019	Baltà-Cruz 2009 (28)	Mueller 2009 (29)	Schneider 2009 (30)	Xynos 2009 (16)	Hakimi 2010 (31)
Table 1.	Patient Number	15	16	17	18	19	20	21 (Fig 2)	22	23	24	25	26

mment	follow-up						nb Finally :ured	nb Finally :ured	esented erous skin ılcers						Continued)
රි	Lost to						L-An	L-An	Promume						Ŭ
Outcome		Cure	Cure	Cure	Cure	Cure	Relapse	Relapse	Cure	Cure	Cure	Cure	Cure	Cure	
Treatment	None	Intralesional MgA	L-AmB 40mg/ Kg Anti-TNF Stopped	Intralesional MgA + Cryotherapy Anti-TNF Stopped	Parenteral Sb Anti-TNF Stopped	L-AmB 40mg/ Kg Anti-TNF Stopped	Intralesional MgA	Intralesional MgA	L-Amb 40mg/ Kg Anti-TNF Stopped	Intralesional MgA Anti-TNF Stopped	Intralesional MgA Anti-TNF Stopped	Intralesional MgA	Intralesional MgA Anti-TNF Stopped	Intralesional MgA Anti-TNF Stopped	
Diagnostic	PCR+	PCR+	DO+ PCR+	DO+ Culture +	PCR+	D0+	PCR+	PCR+	DO+ PCR+ Culture+	D0+	DO+ PCR+	DO+ PCR+	DO+ PCR+	PCR+	
Sample	Skin biopsy	Skin biopsy	Skin biopsy	Skin biopsy	Skin biopsy	Skin biopsy	Skin biopsy	Skin biopsy	Skin biopsy	Skin biopsy	Skin biopsy	Skin biopsy	Skin biopsy	Skin biopsy	
Specie	L. infantum	L. infantum	Leishmania sp.	Leishmania sp.	Leishmania sp.	Leishmania sp.	Leishmania sp.	Leishmania sp.	L. infantum	Leishmania sp.	Leishmania sp.	Leishmania sp.	Leishmania sp.	Leishmania sp.	
Number of lesions	9	1	1	г	-	1	1	1	12	1	1	1	2	1	
Clinical form	CL	CL	CL	C	CL	CL	CL	CL	CL	CL	CL	CL	CL	CL	
Disease	RA	PA	а.	PA	SNA	RA	8	8	RA+P	d	AS	പ	PA	ď	
Sex-Age	F-61	F-59	M-50	M-33	F-45	F-59	M-35	F-55	M-55	F-34	M-46	F-40	M-71	M-53	
Other IS		MTX 20mg/w	1	Leflunomide 20mg/d Pred 5mg/d	XTM	1	I	ı	I	1	1	1	MTX		
Anti TNF	Adalimumab	Adalimumab	Infliximab	Infliximab	Adalimumab	Adalimumab	Adalimumab	Infliximab	Not specified	Adalimumab	Adalimumab	Adalimumab	Adalimumab	Adalimumab	
Country (Region)	Spain (Madrid)	Spain (Madrid)	Spain (Murcia)	Spain (Catalonia)	Malta	Spain (Catalonia)	Spain (Catalonia)	Spain (Catalonia)	Spain (Balearic Islands)	Spain (Valencian Comunity)	Spain (Valencian Comunity)	Spain (Valencian Comunity)	Spain (Valencian Comunity)	Spain (Valencian Comunity)	
Author, year and reference	Romero- Maté 2013 (32)	Romero- Maté 2013 (32)	Hernández- Torres 2013 (33)	Català 2014 (34)	Micallef 2014 (35)	Català 2015 (36)	Marcoval 2017 (37)	Marcoval 2017 (37)	Alcover 2018 (38)	Martínez- Doménech 2019 (39)	Martínez- Doménech 2019 (39)	Martínez- Doménech 2019 (39)	Martínez- Doménech 2019 (39)	Martínez- Doménech 2019(39)	
Patient Number	27	28	29	30	31	32	33	34	35	36	37	38	39	40	

Table 1. (Continued)

Comment		Presented numerous skin ulcers						L-AmB discontinued after 4 doses	
Outcome	Cure	Cure	Cure	Cure	Cure	Cure	Cure	Cure	Curre
Treatment	Intralesional MgA	L-AmB 20mg/ Kg Anti-TNF Stopped	Surgery	Intralesional MgA	L-AmB 20mg/ Kg	L-AmB 24mg/ Kg Anti-TNF Stopped	L-AmB 20mg/ Kg	L-AmB 16mg/ Kg Anti-TNF Stopped	L-AmB 20mg/ Kg Anti-TNF Stopped
Diagnostic	DO+ PCR+	DO+ PCR+	D0+	DO+ PCR +	-D0+	D0+	PCR+	DO+	PCR+
Sample	Skin biopsy	Skin biopsy	Skin biopsy	Skin biopsy	Skin biopsy	Skin biopsy	Skin biopsy	Skin biopsy	Skin biopsy
Specie	Leishmania sp.	L. Infantum	Leishmania sp.	Leishmania sp.	Leishmania sp.	Leishmania sp.	Leishmania sp.	Leishmania sp.	Leishmania sp.
Number of lesions	2	4	1	1	1	-	2	1	5
Clinical form	CL	G	CL	CL	CL	G	CL	C	, Ct
Disease	FD	AS	AS	8	PA	uc	PA	Ð	<u>م</u>
Sex-Age	M-22	M-42	M-67	F-54	M-71	M-67	M-50	M-42	M-74
Other IS		,	Deflazacort 6mg/d	Azatioprine		,		Azatioprine	
Anti TNF	Adalimumab	Infliximab	Infliximab	Infliximab	Golimumab	Infliximab	Infliximab	Infliximab	Adalimumab
Country (Region)	Spain (Valencian Comunity)	Spain (Catalonia)	Spain (Catalonia)	Spain (Catalonia)	Spain (Catalonia)	Spain (Catalonia)	Spain (Catalonia)	Spain (Catalonia)	Spain (Catalonia)
Author, year and reference	Martínez- Doménech 2019(39)	Bosch- Nicolau 2019	Bosch- Nicolau 2019	Bosch- Nicolau 2019	Bosch- Nicolau 2019	Bosch- Nicolau 2019	Bosch- Nicolau 2019	Bosch- Nicolau 2019	Bosch- Nicolau 2019
Patient Number	41	42 (Fig 1)	43	44	45	46	47	48	49

leishmaniasis, MCL: mucucutaneous leishmaniasis, DO: Direct Observation, PCR: Polymerase Chain Reaction, Sb: Sodium tibogluconate, MgA: Meglumine antimoniate; L-AmB: Liposamal Amphotericine B, TNF: Tumor necrosis Factor.

https://doi.org/10.1371/journal.pntd.0007708.t001

PLOS Neglected Tropical Diseases | https://doi.org/10.1371/journal.pntd.0007708 August 30, 2019

Table 1. (Continued)



Fig 1. Patient with multiple cutaneous leishmania lesions. https://doi.org/10.1371/journal.pntd.0007708.g001

In thirty-two (65.3%) cases TNF- α blockers therapy was interrupted. After treatment, four (8.2%) patients with CL diagnosis relapsed. Three of these cases were initially treated with local medication and anti-TNF- α was not stopped. After relapse, the three patients received systemic treatment and anti-TNF- α therapy was discontinued in one patient, all achieving clinical cure. Another relapsing patient was initially treated with miltefosine and finally cured with local antimonials therapy. TNF- α blocker therapy was discontinued in both treatment courses. The last relapsing patient was a VL treated with L-AmB and TNF- α blockers were not stopped as her rheumatic disease was active presenting a MCL form 20 months after. When



Fig 2. Patient with a mucocutaneous leishmaniasis. https://doi.org/10.1371/journal.pntd.0007708.g002



Fig 3. Leishmaniasis form, treatment and outcomes of all included patients. CL: cutaneous leishmaniasis, S: solitary lesion; Mx: multiple lesions; L-AmB: liposomal amphotericine-B; Sb S: systemic sodium tibogluconate; MTF: miltefosine; Mg IL: intralesional meglumine antimoniate; Surg: surgery; TNF: tumor necrosis factor; Ø no treatment, ¥ deceased. Arrows represent patients who relapsed and its path shows patient initial treatment and TNF-alpha blocker management.

https://doi.org/10.1371/journal.pntd.0007708.g003

comparing clinical cure of CL patients, although statistical significance could not be reached cure ratios were 92.3% vs. 78.6% (p = 0.6) when patients received systemic treatment or not and 94.1% vs 70% (p = 0.13) when TNF- α blockers therapy could be stopped or not. Finally, the patient who did not receive any treatment was lost to follow up. Two (4.1%) patients died: one patient after a bacterial superinfection in relation to his immunosuppression and the other patient as a result of a fatal arrhythmia during his treatment with systemic antimonials. Case presentation, treatment and outcomes are represented in Fig 3.

Discussion

According to the published data, TNF- α blockers based therapy seems to be associated with a higher risk of infections, at times with a worse outcome [3]. Nevertheless their efficacy in the management of many chronic inflammatory diseases has reaffirmed its use, representing one of the major breakthroughs in the treatment of these diseases [40].

L. infantum is the major causative agent of leishmaniasis in Southern Europe and the most frequent zymodeme is MON 1 [41]. Even though it has been suggested that there are dermato-tropic and viscerotropic strains, *L. infantum* is prone to produce the visceral form of disease and less frequently the cutaneous form. Although different gens have been proposed to contribute to the viscerlalization process, its mechanisms still remains unclear due to the lack of good animal models [42].

It is also well known that in *L. infantum* endemic areas there is a frequent natural challenge to *Leishmania* parasites, producing circulating parasitemia in the host during an undefined non-permanent time-span. Some authors have referred to this subclinical form as "asymptom-atic carriage". It means that these patients without symptoms have parasite load only detectable by highly sensitive methods such as RT-PCR [43,44].

In southern Europe the prevalence of asymptomatic carriers is remarkably variable (from 0.5% to 48%) depending on the test used for detection [45]. Those data may suggest that there are natural and frequent challenges between host and parasite in endemic areas, and that an effective immune system is required to control clinical expression.

While it seems that in mammalian hosts *Leishmania* species may be able to infect and develop within non-hematopoietic cells such as fibroblasts it is mostly inside monocyte/mac-rophage lineage cells where the parasite replicates and develops long-time survival [46]. Since these cells have a central role in both innate response and acquired immune response as parasite antigen presenting, *Leishmania* interferes in an extremely complex manner with the host immunological response [47].

The role of TNF- α in the pathogenesis of the disease has been widely studied in both humans and animal models. As innate TNF- α dependant mechanisms drive cell mediated immunity by activation of CD4+ and CD8+ T cells, TNF- α is known to be fundamental in the initial control of the infection together with other cytokines such as IL-12 and IFN- γ [48]. These cytokines mediate the development of an effective CD4+ Th1 response which is critical to control the infection [49,50]. Furthermore, TNF- α and IFN- γ are responsible for activating leishmanicidal activity of macrophage which is characterized by an increased production of free oxygen radicals and nitric oxide (NO) as well as inducing infected cells apoptosis [51,52].

In CL, a polarized Th1 response with low Th2 cytokines has been related to infection resistance and disease outcome. Besides, Th1 and Th2 pathways may also modify disease expression [53]. However, in patients presenting disseminated diseases such as VL, immune response seems to be a net suppressive-type response. Rather than an inert Th1 type response, a Th2 / Th1 type response associated with the production of down-regulating cytokines such as IL-10 and TGF- β seems to be responsible of the persistent inflammation, the hallmark of clinical leishmaniasis [54,55]. Moreover, regulatory T- and Th17- cells appear to play an important role in susceptibility and disease resistance [56].

Hence, TNF- α has been implicated in the initial events of the infection, in direct leishmanicidal activity and thus controlling the multiplication of the parasite, developing effective acquired immunity for long term control of the disease as well as mediating the disease expression [47].

The classical VL in Mediterranean countries has been found primarily in children and HIV/AIDS infected patients with poor immune status. Nowadays VL cases in adults are also diagnosed among non-VIH immunosuppressed patients [57]. Moreover, CL is less frequent than VL in some Mediterranean countries as Spain although it may be underreported [58]. Besides, MCL is considered exceptional in the Old World [43]. Anyhow, in our series the proportion of CL and MCL it is surprisingly high.

Clinical outcome observed in this series is unusual. The natural course of CL caused by *L. infantum* is benign and trends to heal spontaneously. In fact, if lesions are less than 5 cm in diameter and are localized in areas where there is no risk of disfiguring or disabling, only a local wound care is recommended [59]. Four out of twenty-eight (14.3%) cases of CL reported in this series relapsed despite etiological treatment.

Despite of the natural viscerotropism of *L. infantum*, the main acquisition of this infection is through the skin. Besides, in endemic areas there is a high exposure to *Leishmania*. Although some studies report up to 48% of prevalence [60], there are only limited cases of CL reported. Consequently, it may be assumed that within immune-competent individuals, the immunity of the host is capable to control the infection, even before that it causes any clinical symptoms.

The reactivation of a latent parasitic form from a granulomatous lesion could be a feasible explanation for those cases. Clinical data from patient number 21 supports this assumption. This patient was diagnosed with a MCL form (Fig 2) 4 years after a CL in his left ankle demonstrated by PCR amplification for *Leishmania* kinetoplastid sequences which healed

spontaneously. Because of the latency between the first episode and the second one and the appearance of a distant lesion far from the initial one, the episode was thus considered as a reactivation in the context of the TNF- α blockers therapy.

Previous series including patients with Old and New world leishmania species showed no consensus regarding treatment in patients under TNF- α blockers [26]. In this series only including leishmaniasis acquired at Mediterranean basin countries, besides all patients with VL and MCL were given a systemic therapy, treatments differ considerably. Seventeen (80.9%) patients received L-AmB at different doses ranging from 25.4mg/Kg to 50mg/Kg divided in 5 to 10 doses, three patients received pentavalent antimonials and one miltefosine. L-AmB is usually recommended as first-choice treatment of VL offering similar cure-rates than pentavalent antimonial [61]. Especially, considering that pentavalent antimonials present a worse toxicity profile and major toxicities such cardiac arrhythmias or prolonged QTc interval that can lead to death as in case number 20 of our series [62]. Although miltefosine has successfully been used treating VL, when it comes to *L. infantum*, a growing number of treatment failures have been published so further investigation is needed before its systematic recommendation [63]. Concerning CL treatment, approach disparities are even greater. Half patients received systemic treatment with different therapies and dose disparity and the other half received local treatment.

As mentioned before, TNF- α plays a key role in the process of controlling infection that ranges from limiting the replication of the parasite up to eliciting an effective adaptative response. TNF- α blockers therapy could modulate the immunological response to a less effective control of the parasite, hence allowing the infection reactivation or more evident disease expression of newly acquired infections. To this end, patients under treatment with TNF- α blockers have less effective immune mediated mechanisms to control and eradicate a parasite challenge, ending in a higher chance to develop classical and non-classical *L. infantum* presenting forms, that otherwise it would have caused a transient parasitemia or an aborted local replication. Likewise, the course and outcome of the disease may be directly affected by the inhibition of TNF- α . Therefore, discontinuation of the anti TNF- α therapy seems to play an important role in the treatment success. Although miltefosine is an excellent option for the New World CL, evidence is scarce for the treatment of *L. infantum* infection [64]. Thus, it seems appropriate to limit its use when other treatment options are not available or have failed.

According to the cases reported, in our opinion the best treatment strategy would be a systemic treatment and the discontinuation of the TNF- α blockers therapy. L-AmB has proved to have the best safety profile and compelling evidence of its efficacy in immunosuppressed patients. One of the major concerns is the reintroduction of TNF- α blockers. Taking into account published information, it seems reasonable reintroducing them once clinical cure has been achieved ensuring close clinical follow-up and blood RT-PCR. Although there is scarce information for its recommendation, etanercep or certolizumab have been suggested as a therapeutical option instead of re-introducing other anti-TNF- α monoclonal antibodies due to its possible lower risk of reactivating leishmaniasis [65–66].

This study has the limitations of any retrospective review study. Although is one of the biggest series published on this topic, the scarce number of patients analyzed limits the strength of the recommendations. We agree with existing reports on the fact that despite of not being a common disease in our area, leishmaniasis is likely underreported [58]. Finally, with the results extracted from this study it is not possible to assess the risk of developing clinical leishmaniasis during anti TNF- α therapy. However, there is enough supporting data for the biological plausibility of the influence of the use of such therapy in the appearance of new cases of clinical leishmaniasis with a switch in the disease expression and outcome. The blockage of TNF- α could determine the incapacity to control and eradicate the parasite within the granuloma playing an important role in increasing the risk of progression to clinical disease.

Conclusions

The increase in the use of TNF- α antagonist has been associated with the emergence of new cases of leishmaniasis. The blockage of TNF- α favors the reactivation of latent leishmaniasis modulating its expression and worsening its clinical outcome.

Once the leishmaniasis is confirmed, systemic drug treatment and the discontinuation of the TNF- α blockers therapy until clinical recovery seems to be the best therapeutic approach when possible.

Otherwise, those patients receiving such therapy and coming from endemic areas require a close monitoring in order to detect early forms and start adequate treatment.

Prospective studies and more participation on declaring existing cases in the adverse events notification system is required in order to assess the risk of leishmaniasis and other opportunistic diseases related to the use of anti TNF- α treatment more accurately.

Supporting information

S1 Checklist. STROBE checklist. (DOCX)

Author Contributions

Conceptualization: Pau Bosch-Nicolau, Fernando Salvador, Adrián Sánchez-Montalvá, Israel Molina.

Formal analysis: Pau Bosch-Nicolau, Israel Molina.

Investigation: Pau Bosch-Nicolau, Maria Ubals, Fernando Salvador, Adrián Sánchez-Montalvá.

Project administration: Israel Molina.

Supervision: Gloria Aparicio, Alba Erra, Pablo Martinez de Salazar, Elena Sulleiro.

Writing - original draft: Pau Bosch-Nicolau.

Writing – review & editing: Maria Ubals, Fernando Salvador, Adrián Sánchez-Montalvá, Elena Sulleiro, Israel Molina.

References

- 1. Wallis RS, Broder M, Wong J, Lee A, Hoq L. Reactivation of Latent Granulomatous Infections by Infliximab. Clin Infect Dis. 2005 Aug 1; 41(Supplement_3):S194–8.
- Lis K, Kuzawińska O, Bałkowiec-Iskra E. State of the art paper Tumor necrosis factor inhibitors-state of knowledge. Arch Med Sci. 2014 Dec 22; 6(6):1175–85.
- Pérez-Sola MJ, Torre-Cisneros J, Pérez-Zafrilla B, Carmona L, Descalzo MA, Gómez-Reino JJ, et al. Infections in patients treated with tumor necrosis factor antagonists: incidence, etiology and mortality in the BIOBADASER registry. Med Clin (Barc). 2011 Nov 12; 137(12):533–40.
- Burza S, Croft SL, Boelaert M. Leishmaniasis. Lancet. 2018 Sep 15; 392(10151):951–70. https://doi. org/10.1016/S0140-6736(18)31204-2
- Guedes-Barbosa LS, Pereira da Costa I, Fernandes V, Henrique da Mota LM, de Menezes I, Aaron Scheinberg M. Leishmaniasis during anti-tumor necrosis factor therapy: Report of 4 cases and review of the literature (additional 28 cases). Semin Arthritis Rheum. 2013 Oct; 43(2):152–7. https://doi.org/10. 1016/j.semarthrit.2013.01.006 PMID: 23777708
- Zanger P, Kötter I, Kremsner PG, Gabrysch S. Tumor necrosis factor alpha antagonist drugs and leishmaniasis in Europe. Clin Microbiol Infect. 2012 Jul; 18(7):670–6. <u>https://doi.org/10.1111/j.1469-0691.</u> 2011.03674.x PMID: 21985586

- Molina I, Fisa R, Riera C, Falcó V, Elizalde A, Salvador F, et al. Ultrasensitive real-time PCR for the clinical management of visceral leishmaniasis in HIV-Infected patients. Am J Trop Med Hyg. 2013 Jul 10; 89(1):105–10. https://doi.org/10.4269/ajtmh.12-0527 PMID: 23629932
- Romaní-Costa V, Sánchez C, Moyá F, Estany C. Visceral leishmaniasis related to infliximab administration. Enferm Infecc Microbiol Clin. 2004 May; 22(5):310.
- Fabre S, Gibert C, Lechiche C, Dereure J, Jorgensen C, Sany J. Visceral leishmaniasis infection in a rheumatoid arthritis patient treated with infliximab. Clin Exp Rheumatol. 2005 Nov-Dec; 23(6):891–2. PMID: 16396712
- Bagalas V, Kioumis I, Argyropoulou P, Patakas D. Visceral leishmaniasis infection in a patient with rheumatoid arthritis treated with etanercept. Clin Rheumatol. 2007 Aug 27; 26(8):1344–5. <u>https://doi.org/10.1007/s10067-006-0356-5 PMID: 16865310</u>
- Bassetti M, Pizzorni C, Gradoni L, Del Bono V, Cutolo M, Viscoli C. Visceral leishmaniasis infection in a rheumatoid arthritis patient treated with adalimumab. Rheumatology. 2006 Nov; 45(11):1446–8. <u>https:// doi.org/10.1093/rheumatology/kel235 PMID: 16887847</u>
- Tektonidou MG, Skopouli FN. Visceral leishmaniasis in a patient with psoriatic arthritis treated with infliximab: reactivation of a latent infection? Clin Rheumatol. 2008 Apr 26; 27(4):541–2. <u>https://doi.org/ 10.1007/s10067-007-0775-y PMID: 17963018</u>
- De Leonardis F, Govoni M, Lo Monaco A, Trotta F. Visceral leishmaniasis and anti-TNF-alpha therapy: case report and review of the literature. Clin Exp Rheumatol. 2009 May-Jun; 27(3):503–6. PMID: 19604446
- Garcia-Vidal C, Rodríguez-Fernández S, Teijón S, Esteve M, Rodríguez-Carballeira M, Lacasa JM, et al. Risk factors for opportunistic infections in infliximab-treated patients: the importance of screening in prevention. Eur J Clin Microbiol Infect Dis. 2009 Apr 17; 28(4):331–7. <u>https://doi.org/10.1007/s10096-008-0628-x</u> PMID: 18797940
- Jeziorski E, Blanchet C, Ludwig C, Lalande M, Coste V, Dereure J, et al. Récidive sous forme pseudotumorale d'une leishmaniose viscérale. Arch Pédiatrie. 2009 Oct; 16:S129–31.
- Xynos ID, Tektonidou MG, Pikazis D, Sipsas NV. Leishmaniasis, autoimmune rheumatic disease, and anti-tumor necrosis factor therapy, Europe. Emerg Infect Dis. 2009 Jun; 15(6):956–9. <u>https://doi.org/10.3201/eid1506.090101</u> PMID: 19523302
- Moreno D, Martínez P, Berbegal J, Femenia M. [Visceral leishmaniasis infection in a rheumatoid arthritis patient treated with adalimumab: a case description and literature review]. Enferm Infecc Microbiol Clin. 2010 Apr; 28(4):261–2. https://doi.org/10.1016/j.eimc.2009.03.017 PMID: 19729230
- Moltó A, Mateo L, Lloveras N, Olivé A, Minguez S. Visceral leishmaniasis and macrophagic activation syndrome in a patient with rheumatoid arthritis under treatment with adalimumab. Joint Bone Spine. 2010 May; 77(3):271–3. https://doi.org/10.1016/j.jbspin.2010.01.011 PMID: 20378385
- Kritikos K, Haritatos E, Tsigkos S, Gounari P, Skrapari I, Gounaris T, et al. An atypical presentation of visceral leishmaniasis infection in a patient with rheumatoid arthritis treated with infliximab. J Clin Rheumatol. 2010 Jan; 16(1):38–9. https://doi.org/10.1097/RHU.0b013e3181c8aa33 PMID: 20051756
- Erre GL, Mesina P, Tonelli N, Passiu G. Visceral Leishmaniasis among immunosuppressed patients with rheumatic diseases. Clin Exp Rheumatol.; 28(4):590–1. PMID: 20659417
- Khan A, Coakley G, Cosgrove C, Lockwood D. Let off the leash: kala-azar following the use of tumour necrosis factor antibodies. Case Reports. 2010 Oct 13; 2010(oct12 2): bcr0420102878.
- 22. Besada E, Njålla RJ, Nossent JC. Imported case of visceral leishmaniasis presenting as pancytopenia in a Norwegian patient treated with methotrexate and etanercept for psoriasis arthritis. Rheumatol Int. 2013 Oct 12; 33(10):2687–9. https://doi.org/10.1007/s00296-012-2483-4 PMID: 22886470
- 23. Berry CE, Tsai J, Tierney A, Pickles R. Visceral leishmaniasis in a patient taking adalimumab for rheumatoid arthritis. Med J Aust. 2013 Apr 1; 198(6):331–3. PMID: 23545033
- García-Castro JM, Javier-Martínez R, Cáliz-Cáliz R, García-Sánchez A. Persistent inflammation of the nasal dorsum in a patient with rheumatoid arthritis treated with infliximab. Enferm Infecc Microbiol Clin. 2012 Jan; 30(1):40–2. https://doi.org/10.1016/j.eimc.2011.06.010 PMID: 21903302
- Garcia-Gonzalez E, Guidelli GM, Bardelli M, Maggio R. Mucocutaneous leishmaniasis in a patient treated with anti-TNF- therapy. Rheumatology. 2012 Aug 1; 51(8):1517–8. https://doi.org/10.1093/ rheumatology/kes030 PMID: 22427411
- 26. Neumayr ALC, Morizot G, Visser LG, Lockwood DNJ, Beck BR, Schneider S, et al. Clinical aspects and management of cutaneous leishmaniasis in rheumatoid patients treated with TNF-α antagonists. Travel Med Infect Dis. 2013 Nov; 11(6):412–20. https://doi.org/10.1016/j.tmaid.2013.05.003 PMID: 23972958
- Juzlova K, Votrubova J, Kacerovska D, Lukas M, Bortlik M, Rohacova H, et al. Visceral leishmaniasis with cutaneous symptoms in a patient treated with infliximab followed by fatal consequences. Dermatol Ther. 2014 May; 27(3):131–4. https://doi.org/10.1111/dth.12083 PMID: 24903470

- Baltà-Cruz S, Alsina-Glbert M, Mozos-Rocafort A, Cervera C, Colomo-Saperas L, Del Río A, et al. Pseudolymphomatoid cutaneous leishmaniasis in a patient treated with adalimumab for rheumatoid arthritis. Acta Derm Venereol. 2009; 89(4):432–3. https://doi.org/10.2340/00015555-0633 PMID: 19688168
- Mueller MC, Fleischmann E, Grunke M, Schewe S, Bogner JR, Löscher T. Relapsing cutaneous leishmaniasis in a patient with ankylosing spondylitis treated with infliximab. Am J Trop Med Hyg. 2009 Jul; 81(1):52–4. PMID: 19556566
- Schneider P, Bouaziz JD, Foulet F, Duong TA, Allanore LV, Bagot M. Multifocal cutaneous leishmaniasis due to Leishmania infantum under adalimumab therapy. Ann Dermatol Venereol. 2009 Nov; 136 (11):815–6. https://doi.org/10.1016/j.annder.2009.05.008 PMID: 19917436
- Hakimi S, Rivière S, Del Giudice P, Dereure J, Le Quellec A. Localized Cutaneous Leishmaniasis due to *Leishmania infantum* in a Patient Treated with Infliximab. Dermatology. 2010; 220(1):63–5. <u>https:// doi.org/10.1159/000258675 PMID: 19907139</u>
- Romero-Maté A, Martínez-Sánchez D, Tardío JC, Moreno-Torres A, García-Donoso C, Aguado-Lobo M, et al. Cutaneous leishmaniasis with histopathologic pattern of non-necrotizing granulomatous dermatitis in patients treated with adalimumab. Dermatol Online J. 2012 Sep 15; 18(9):7. PMID: 23031374
- Hernández-Torres A, García-Vázquez E, Frías-Iniesta J, Herrero-Martínez JA, Gómez-Gómez J. Cutaneous leishmaniasis in a patient receiving infliximab. Scand J Infect Dis. 2013 Jul 28; 45(7):567–9. https://doi.org/10.3109/00365548.2012.752859 PMID: 23270475
- Català A, Barnadas MA, Muñoz C, Puig L. Cutaneous leishmaniasis in a patient receiving infliximab for psoriatic arthritis: treatment with cryotherapy and intralesional meglumine antimonate. Actas Dermosifiliogr. 2014 Sep; 105(7):714–6. https://doi.org/10.1016/j.ad.2013.08.011 PMID: 24275564
- 35. Micallef C, Azzopardi CM. Atypical cutaneous leishmaniasis in the immunosuppressed. BMJ Case Rep. 2014 Jun 10; 2014(jun10 1):bcr2014204914-bcr2014204914.
- Català A, Roé E, Dalmau J, Pomar V, Muñoz C, Yelamos O, et al. Anti-tumour necrosis factor-induced visceral and cutaneous leishmaniasis: case report and review of the literature. Dermatology. 2015; 230 (3):204–7. https://doi.org/10.1159/000370238 PMID: 25633623
- Marcoval J, Penín RM, Sabé N, Valentí-Medina F, Bonfill-Ortí M, Martínez-Molina L. Cutaneous leishmaniasis associated with anti-tumour necrosis factor-α drugs: an emerging disease. Clin Exp Dermatol. 2017 Apr; 42(3):331–4. https://doi.org/10.1111/ced.13061 PMID: 28239885
- Alcover MM, Rocamora V, Guillén MC, Berenguer D, Cuadrado M, Riera C, et al. Case Report: Diffuse Cutaneous Leishmaniasis by Leishmania infantum in a Patient Undergoing Immunosuppressive Therapy: Risk Status in an Endemic Mediterranean Area. Am J Trop Med Hyg. 2018 May 9; 98(5):1313–6. https://doi.org/10.4269/ajtmh.17-0769 PMID: 29582737
- Martínez-Doménech A, García-Legaz-Martínez M, Magdaleno-Tapial J, Torrecillas M, Hernández-Bel P, Ocete Mochón MD, et al. Anti-TNF-associated cutaneous leishmaniasis: a single-institution experience. Br J Dermatol. 2019 Jan 19; https://doi.org/10.1111/bjd.17644 PMID: 30659588
- Steeland S, Libert C, Vandenbroucke RE. A New Venue of TNF Targeting. Int J Mol Sci. 2018 May 11; 19(5):1442.
- Moriconi M, Rugna G, Calzolari M, Bellini R, Albieri A, Angelini P, et al. Phlebotomine sand fly-borne pathogens in the Mediterranean Basin: Human leishmaniasis and phlebovirus infections. PLoS Negl Trop Dis. Public Library of Science; 2017 Aug; 11(8):e0005660.
- 42. Nafchi HR, Kazemi-Rad E, Mohebali M, Raoofian R, Ahmadpour NB, Oshaghi MA, et al. Expression analysis of viscerotropic leishmaniasis gene in Leishmania species by real-time RT-PCR. Acta Parasitol. 2016 Jan 1; 61(1):93–7. https://doi.org/10.1515/ap-2016-0011 PMID: 26751877
- 43. Riera C, Fisa R, Lpez-Chejade P, Serra T, Girona E, Jimnez M, et al. Asymptomatic infection by *Leishmania infantum* in blood donors from the Balearic Islands (Spain). Transfusion. 2008 Jul; 48(7):1383–9. https://doi.org/10.1111/j.1537-2995.2008.01708.x PMID: 18422844
- 44. Foroutan M, Dalvand S, Khademvatan S, Majidiani H, Khalkhali H, Masoumifard S, et al. A systematic review and meta-analysis of the prevalence of Leishmania infection in blood donors. Transfus Apher Sci. 2017 Aug; 56(4):544–51. https://doi.org/10.1016/j.transci.2017.07.001 PMID: 28754214
- Michel G, Pomares C, Ferrua B, Marty P. Importance of worldwide asymptomatic carriers of Leishmania infantum (L. chagasi) in human. Acta Trop. 2011 Aug; 119(2–3):69–75. https://doi.org/10.1016/j. actatropica.2011.05.012 PMID: 21679680
- 46. Bogdan C, Donhauser N, Döring R, Röllinghoff M, Diefenbach A, Rittig MG. Fibroblasts as host cells in latent leishmaniosis. J Exp Med. 2000 Jun 19; 191(12):2121–30. <u>https://doi.org/10.1084/jem.191.12</u>. 2121 PMID: 10859337
- Kaye P, Scott P. Leishmaniasis: complexity at the host-pathogen interface. Nat Rev Microbiol. 2011 Jul 11; 9(8):604–15. https://doi.org/10.1038/nrmicro2608 PMID: 21747391

- Gannavaram S, Bhattacharya P, Ismail N, Kaul A, Singh R, Nakhasi HL. Modulation of Innate Immune Mechanisms to Enhance Leishmania Vaccine-Induced Immunity: Role of Coinhibitory Molecules. Front Immunol. 2016 May 13; 7:187. https://doi.org/10.3389/fimmu.2016.00187 PMID: 27242794
- von Stebut E, Udey MC. Requirements for Th1-dependent immunity against infection with Leishmania major. Microbes Infect. 2004 Oct; 6(12):1102–9. https://doi.org/10.1016/j.micinf.2004.05.024 PMID: 15380780
- Engwerda CR, Ato M, Kaye PM. Macrophages, pathology and parasite persistence in experimental visceral leishmaniasis. Trends Parasitol. 2004 Nov; 20(11):524–30. https://doi.org/10.1016/j.pt.2004.08. 009 PMID: 15471704
- Souza AS, Giudice A, Pereira JM, Guimarães LH, de Jesus AR, de Moura TR, et al. Resistance of Leishmania (Viannia) braziliensis to nitric oxide: correlation with antimony therapy and TNF-alpha production. BMC Infect Dis. 2010 Jul 15; 10(1):209.
- Allenbach C, Zufferey C, Perez C, Launois P, Mueller C, Tacchini-Cottier F. Macrophages induce neutrophil apoptosis through membrane TNF, a process amplified by Leishmania major. J Immunol. 2006 Jun 1; 176(11):6656–64. https://doi.org/10.4049/jimmunol.176.11.6656 PMID: 16709824
- Maspi N, Abdoli A, Ghaffarifar F. Pro- and anti-inflammatory cytokines in cutaneous leishmaniasis: a review. Pathog Glob Health. 2016 Aug 17; 110(6):247–60. <u>https://doi.org/10.1080/20477724.2016</u>. 1232042 PMID: 27660895
- Stanley AC, Engwerda CR. Balancing immunity and pathology in visceral leishmaniasis. Immunol Cell Biol. 2007 Feb; 85(2):138–47. https://doi.org/10.1038/sj.icb7100011 PMID: 17146466
- Sacks D, Anderson C. Re-examination of the immunosuppressive mechanisms mediating non-cure of Leishmania infection in mice. Immunol Rev. 2004 Oct; 201(1):225–38.
- 56. Gonçalves-de-Albuquerque S da C, Pessoa-e-Silva R, Trajano-Silva LAM, de Goes TC, de Morais RCS, da C. Oliveira CN, et al. The Equivocal Role of Th17 Cells and Neutrophils on Immunopathogenesis of Leishmaniasis. Front Immunol. 2017 Oct 30; 8:1437. <u>https://doi.org/10.3389/fimmu.2017.01437</u> PMID: 29163510
- Monge-Maillo B, Norman FF, Cruz I, Alvar J, López-Vélez R. Visceral Leishmaniasis and HIV Coinfection in the Mediterranean Region. Valenzuela JG, editor. PLoS Negl Trop Dis. 2014 Aug 21; 8(8):e3021. https://doi.org/10.1371/journal.pntd.0003021 PMID: 25144380
- Alvar J, Vélez ID, Bern C, Herrero M, Desjeux P, Cano J, et al. Leishmaniasis Worldwide and Global Estimates of Its Incidence. Kirk M, editor. PLoS One. 2012 May 31; 7(5):e35671. <u>https://doi.org/10.1371/journal.pone.0035671</u> PMID: 22693548
- Aronson N, Herwaldt BL, Libman M, Pearson R, Lopez-Velez R, Weina P, et al. Diagnosis and Treatment of Leishmaniasis: Clinical Practice Guidelines by the Infectious Diseases Society of America (IDSA) and the American Society of Tropical Medicine and Hygiene (ASTMH). Am J Trop Med Hyg. 2017 Jan 11; 96(1):24–45. https://doi.org/10.4269/ajtmh.16-84256 PMID: 27927991
- Mary C, Faraut F, Drogoul M-P, Xeridat B, Schleinitz N, Cuisenier B, et al. Reference values for Leishmania infantum parasitemia in different clinical presentations: quantitative polymerase chain reaction for therapeutic monitoring and patient follow-up. Am J Trop Med Hyg. 2006 Nov]; 75(5):858–63. PMID: 17123977
- Pagliano P, Esposito S. Visceral leishmaniosis in immunocompromised host: an update and literature review. J Chemother. 2017 Sep 3; 29(5):261–6. <u>https://doi.org/10.1080/1120009X.2017.1323150</u> PMID: 28490252
- Sundar S, Chakravarty J. An Update on pharmacotherapy for Leishmaniasis. Expert Opin Pharmacother. 2015 Feb; 16(2):237–52. https://doi.org/10.1517/14656566.2015.973850 PMID: 25346016
- Hendrickx S, Beyers J, Mondelaers A, Eberhardt E, Lachaud L, Delputte P, et al. Evidence of a drugspecific impact of experimentally selected paromomycin and miltefosine resistance on parasite fitness in Leishmania infantum. J Antimicrob Chemother. 2016 Jul; 71(7):1914–21. <u>https://doi.org/10.1093/jac/</u> dkw096 PMID: 27084919
- 64. Mosimann V, Blazek C, Grob H, Chaney M, Neumayr A, Blum J. Miltefosine for Mucosal and Complicated Cutaneous Old World Leishmaniasis: A Case Series and Review of the Literature. Open forum Infect Dis. Oxford University Press; 2016 Jan; 3(1):ofw008.
- Zanger P, Gabrysch S. Leishmaniasis in the era of tumor necrosis factor alpha antagonist therapy—a research agenda for Europe. Euro Surveill. 2013 Jul 25; 18(30):20542. https://doi.org/10.2807/1560-7917.es2013.18.30.20542 PMID: 23929182
- 66. Arens K, Filippis C, Kleinfelder H, Goetzee A, Reichmann G, Crauwels P, et al. Anti-Tumor Necrosis Factor α Therapeutics Differentially Affect Leishmania Infection of Human Macrophages. Front Immunol. Frontiers Media SA; 2018; 9:1772.